Author’s response to reviews

Title: Scrotal Calcinosis, ethiopathogenic and therapeutic aspects

Authors:

Omar Karray (omar.karray.88@gmail.com)
Amen Ghozzi Dhaoui (amendhaoui@gmail.com)
Rami Boulma (rami_boulma@yahoo.fr)
Hassen Khouni (hass_ElKhouni@gmail.com)
Khadija Bellil (kadija.Bellil@rns.tn)

Version: 2 Date: 02 Sep 2017

Author’s response to reviews:

#Reviewer 1:

The psychological and the familiar context of the patients have been evoked in the revised manuscript (mentioned in italic characters).

As the ethiopathogenesis is still controversial, we have exposed the most recognized theories on this point, with the results of the largest studies of Song et al. and Dubey et al. The theory of dystrohy of epidermal cyst is the most convincing one when reviewing the literature.

# Reviewer 2:

We added two figures of operative specimen for both patient in order to discern the presence of calcified nodules. Pustules couldn't be excised surgically in such a manner. In addition, the figure of histology examination confirm that it's about epidermal caldified lesions. The related revision in the manuscript is mentioned in italic characters.
The manuscript has been revised by the head of the Medical English department in the faculty, in the aim of improving the manuscript quality and level (mentioned in red characters in the manuscript).